Author's response to reviews

Title: Paired design study by real-time PCR: miR-378* and miR-145 are potent early diagnostic biomarkers of human colorectal cancer

Authors:

peng juan (dielianhuayaya@163.com)
xie zhengyong (pwkxiezy@163.com)
cheng liyang (chliyang2008@sina.com)
zhang yuxin (zhangyuxin-53825@163.com)
yu hongping (yhp268@163.com)
chen juyong (294821766@qq.com)
kang huixin (1367706921@qq.com)

Version: 8 Date: 9 January 2015

Author's response to reviews: see over
1. We calculate the power of t-test and ROC curve and it’s more than 0.75 which is enough to find the difference. For the logistic regression, we calculate two variable selection method’s result, which is also quite powerful to find the difference. The results add an another microRNA, miR-145.

2. Part Language correction has been made.